Author's response to reviews

Title: The pro-apoptotic K-Ras 4A proto-oncoprotein does not affect tumorigenesis in the ApcMin/+ mouse small intestine

Authors:

Charles E Patek (charles.patek@hotmail.com)
Mark J Arends (mja40@cam.ac.uk)
Lorraine Rose (lrose@staffmail.ed.ac.uk)
Feijun Luo (fl239@cam.ac.uk)
Marion Walker (Marion.Walker@ed.ac.uk)
Paul S Devenney (pauldev@hgu.mrc.ac.uk)
Rachel L Berry (rberry@hgu.mrc.ac.uk)
Nicola J Lawrence (n.lawrence@erasmusmc.nl)
Rachel A Ridgway (r.ridgway@beatson.gla.ac.uk)
Owen J Sansom (o.sansom@beatson.gla.ac.uk)
Martin L Hooper (m.hooper@ed.ac.uk)

Version: 3 Date: 8 June 2008

Author's response to reviews:

Sir Alastair Currie Cancer Research UK Laboratories
Molecular Medicine Centre
The University of Edinburgh
Western General Hospital
Crewe Road, Edinburgh EH4 2XU, UK.

BioMed Central Editorial Production Team
corepreaccept@biomedcentral.com

Dear Sirs

Manuscript 1922922811171449

Thank you for your Email of 2 June. Please find herewith a reformatted manuscript with the following changes:

Each affiliation is on a new line.

Running title and keywords have been removed.

Present address is changed to “Current address” and each address is on a new line.

Apostrophe has been moved in “Authors’ contributions” and a statement included that all authors read and approved the final manuscript.
LEGENDS has been changed to “Figure Legends”.

I will check figure orientations after uploading. The figures had already been cropped according to your instructions.

I trust that the reformatted manuscript is satisfactory.

Yours sincerely

Martin Hooper